## communications medicine

### ARTICLE

https://doi.org/10.1038/s43856-022-00079-7

OPEN

# Aerobic exercise improves episodic memory in late adulthood: a systematic review and meta-analysis

Sarah L. Aghjayan<sup>® 1,2™</sup>, Themistokles Bournias<sup>3</sup>, Chaeryon Kang<sup>4</sup>, Xueping Zhou<sup>4</sup>, Chelsea M. Stillman<sup>1</sup>, Shannon D. Donofry<sup>1</sup>, Thomas W. Kamarck<sup>1</sup>, Anna L. Marsland<sup>1</sup>, Michelle W. Voss<sup>5</sup>, Scott H. Fraundorf<sup>® 1</sup> & Kirk I. Erickson<sup>1,2,6,7</sup>

#### Abstract

**Background** Aerobic exercise remains one of the most promising approaches for enhancing cognitive function in late adulthood, yet its potential positive effects on episodic memory remain poorly understood and a matter of intense debate. Prior meta-analyses have reported minimal improvements in episodic memory following aerobic exercise but have been limited by restrictive inclusion criteria and infrequent examination of exercise parameters.

**Methods** We conducted a meta-analysis of randomized controlled trials to determine if aerobic exercise influences episodic memory in late adulthood (M = 70.82 years) and examine possible moderators. Thirty-six studies met inclusion criteria, representing data from 2750 participants.

**Results** Here we show that aerobic exercise interventions are effective at improving episodic memory (Hedges'g = 0.28; p = 0.002). Subgroup analyses revealed a moderating effect of age (p = 0.027), with a significant effect for studies with a mean age between 55–68 but not 69–85. Mixed-effects analyses demonstrated a positive effect on episodic memory among studies with a high percentage of females (65–100%), participants with normal cognition, studies reporting intensity, studies with a no-contact or nonaerobic physical activity control group, and studies prescribing >3900 total minutes of activity (range 540–8190 min).

**Conclusions** Aerobic exercise positively influences episodic memory among adults  $\geq$ 55 years without dementia, with larger effects observed among various sample and intervention characteristics—the clearest moderator being age. These results could have far-reaching clinical and public health relevance, highlighting aerobic exercise as an accessible, non-pharmaceutical intervention to improve episodic memory in late adulthood.

#### **Plain Language Summary**

One of the earliest memory systems to decline with increasing age is episodic memory. Episodic memory is the remembrance of past personal events and experiences. Given a lack of pharmaceutical treatments to prevent or reverse this deterioration, it is important to investigate nonpharmaceutical methods for promoting the integrity of episodic memory. It remains unclear whether aerobic exercise induces changes in episodic memory. This study examined the effects of aerobic exercise randomized controlled trials on episodic memory in older adults without dementia and assessed whether the effects depend on the characteristics of the sample and intervention. We found that aerobic exercise positively influences episodic memory, with larger effects observed among various sample and intervention characteristics. These results highlight aerobic exercise as an accessible, non-pharmaceutical intervention to improve episodic memory in late adulthood.

Check for updates

<sup>&</sup>lt;sup>1</sup> Department of Psychology, University of Pittsburgh, Pittsburgh, PA, USA. <sup>2</sup> Center for the Neural Basis of Cognition, University of Pittsburgh, Pittsburgh, PA, USA. <sup>3</sup> Department of Psychology, Carnegie Mellon University, Pittsburgh, PA, USA. <sup>4</sup> Department of Biostatistics, University of Pittsburgh, Pittsburgh, PA, USA. <sup>5</sup> Department of Psychological and Brain Sciences, University of Iowa, Iowa City, IA, USA. <sup>6</sup> Exercise Science, College of Science, Health, Engineering and Education, Murdoch University, Perth, WA, Australia. <sup>7</sup> PROFITH "PROmoting FITness and Health Through Physical Activity" Research Group, Sport and Health University Research Institute (iMUDS), Department of Physical and Sports Education, Faculty of Sport Sciences, University of Granada, 18071 Granada, Spain.

ne of the earliest cognitive domains to decline with increasing age and with Alzheimer's disease (AD) is episodic memory  $(EM)^{1,2}$ . Given that ~6.2 million adults  $\geq$ 65 years in the United States have AD<sup>3</sup> and the population of adults  $\geq 65$  years is projected to reach 88 million by 2050<sup>4</sup>, it is increasingly important to study EM. EM is the remembrance of past personal events and experiences, and it is supported by a distributed network of cortical and subcortical brain structures, including the involvement of the hippocampus and the prefrontal cortex<sup>1,5,6</sup>. A decline in EM is associated with a decrease in the ability to perform activities of daily living and an increase in social isolation<sup>7,8</sup>. Memory complaints often begin to occur even in midlife, with about one in nine adults  $\geq$ 45 years reporting memory problems<sup>9</sup>. EM declines early in the aging trajectory because the hippocampus is more susceptible to agerelated deterioration than other brain regions<sup>10</sup>. Greater hippocampal volume is important because reduced hippocampal volume is associated with conversion to AD over a 2.4-year follow-up among older adults with mild cognitive impairment (MCI), the transitional phase between normal cognition (NC) and dementia, and among older adults with subjective cognitive decline (SCD; i.e., self-reported memory problems)<sup>11,12</sup>. Given a lack of pharmaceutical treatments to prevent or reverse this deterioration, it is important to investigate non-pharmaceutical methods for promoting the integrity of the hippocampus and EM with increasing age.

Fortunately, participation in aerobic exercise (AE) improves markers of brain health that otherwise decline with age and that are associated with an increased risk for AD<sup>13</sup>. AE is a form of physical activity that is structured, planned, repetitive, and performed at an intensity that maintains or improves cardiorespiratory fitness<sup>14</sup>. Cross-sectional, longitudinal, and randomized controlled trials (RCTs) in older adults demonstrate that AE increases hippocampal gray matter volume<sup>15</sup>, hippocampal blood volume<sup>16</sup>, and hippocampal functional connectivity<sup>17</sup>. AE also strengthens functional connectivity of the hippocampus to the default mode network, a hippocampal-cortical brain network that spans the medial and lateral surfaces of the prefrontal, parietal, and temporal cortices<sup>17</sup>. Further, the default network shows reduced functional connectivity with aging and predicts later cognitive impairment<sup>18,19</sup>. These data suggest that regular exercise may be a low-cost, scalable, and highly accessible treatment to either enhance hippocampal-related function or prevent/delay the risk for hippocampal degeneration typical of aging and early stages of pathological cognitive decline.

Despite the distinct benefits of AE on the hippocampus<sup>20</sup>, it remains unclear whether AE induces changes in the cognitive functions supported by the hippocampus, namely EM. Although rodent studies provide irrefutable evidence that AE enhances hippocampal-related memory task performance, the results are more equivocal in humans<sup>21,22</sup>. While some RCTs have found significant improvements in EM among older adults with and without cognitive impairment<sup>23–25</sup>, others have failed to find an effect<sup>26,27</sup>. Several meta-analyses have summarized the findings from these RCTs and found no benefits of AE on EM among older adults with or without cognitive decline<sup>28–33</sup>.

Several limitations of the aforementioned meta-analyses qualify and complicate the interpretation of these null results. First, prior meta-analyses used restricted inclusion criteria, limiting the analyses to studies of either only individuals with NC or only individuals with MCI. By examining each group in isolation, we may miss important information about the trajectory of cognitive aging that would help determine when exercise could be most beneficial. Second, the aforementioned reviews offer little information about the optimal dose of AE, such as intervention length, for positively influencing EM in particular. Determining the doses necessary for improving EM has both theoretical and practical value, as it could help inform mechanistic explanations as well as future interventions and health professionals about levels of AE to prescribe. Additionally, sample characteristics, such as sex distribution, were often overlooked in previous analyses. Numerous RCTs have shown that AE elicits greater cognitive benefits in females than in males<sup>34,35</sup>. Further, females tend to outperform males on tests of EM, and males experience greater incidence rates of EM decline than females<sup>36,37</sup>. Thus, pooling studies with different sample and intervention characteristics may increase heterogeneity and mask significant effects. Third, some of the aforementioned reviews examine the broader domain of memory, which often includes working memory tasks and other tasks not specific to EM. Lastly, 15 relevant studies have been published since the last meta-analysis<sup>30</sup>. Consequently, there is a need for a more complete summary of the available evidence of older adults without dementia, with a focus on the impact of sample and intervention characteristics.

To address the gaps in previous meta-analyses, the current meta-analysis examined dose parameters as potential moderators of the effect of AE on EM and included studies of all adults with a mean age  $\geq$ 55 years without dementia, encompassing the trajectory of cognitive aging from NC to MCI. Thus, this study addresses the following two aims: (1) examine the effects of AE RCTs on EM in adults  $\geq$ 55 years without dementia; and (2) assess whether any of the following characteristics of the sample and intervention moderate the effect of AE on EM: sex distribution, cognitive status, age, control group type, reporting of prescribed intensity, intervention length, session duration, session frequency, and intervention volume. We predicted that AE would improve EM and that this effect would be greater in NC, females, younger ages, interventions with a no-contact control group, and interventions with greater amounts of prescribed exercise. Our results show that AE positively influences EM, with larger effects observed among various sample and intervention characteristics. These results have far-reaching clinical and public health relevance, highlighting the importance of implementing AE early on in the aging process to help mitigate deficits in EM apparent in healthy aging, AD, and other neurodegenerative diseases, such as Parkinson's disease and Huntington's disease.

#### Methods

We conducted this meta-analysis in accordance with established guidelines from Preferred Reporting Items for Systematic Reviews and Meta-Analysis (PRISMA)<sup>38</sup>. Before conducting the search, we registered the proposed meta-analysis on PROSPERO (Registration number: CRD42020222666).

Inclusion and exclusion criteria. Studies were included if they met the following criteria (see Table 1): (1) Studies of males and/ or females with a mean age of 55 years or older. All studies of participants with NC, SCD, and MCI were included, regardless of the diagnostic criteria used. Studies that included clinical samples with other neurological or mental illnesses (e.g., dementia, stroke, depression) were excluded; (2) RCTs of AE (including aerobic dance and aerobic tai chi interventions). Studies that included strength training, exergaming, or cognitive training as part of the AE group were excluded because of their potential confounding impact on cognition through nonaerobic mechanisms. One study that produced three publications described their methods as including some sessions in the anaerobic zone<sup>39,40</sup>. These publications were included here because the majority of the duration of the intervention was conducted in the aerobic zone; (3) Control group type could include no contact, wait-list, stretching, toning, balance, light resistance, education, or social interaction.

Table T Inclusion criteria.										
Parameter	Inclusion criteria	Exclusion criteria								
Participants	Adults with a mean age of 55 years or older with normal cognition, subjective cognitive decline, or mild cognitive impairment	Clinical samples (e.g., depression, dementia)								
Intervention	Randomized controlled trial of an aerobic exercise intervention	Strength training, exergaming, or cognitive training in the aerobic group								
Control	No contact, wait-list, stretching, toning, balance, education, or social interaction	Strength training, exergaming, or cognitive training								
Outcomes	Episodic memory measure, including word list, story recall, face/name recognition, object memory, or paired associates	Composite score that did not exclusively assess episodic memory								
Study design	Any length, frequency, duration, volume, or intensity	Did not isolate effects of aerobic group from control group								

Studies with strength training, exergaming, or cognitive training in the control group were excluded; (4) EM measure consisting of word list, story recall, face/name recognition, object memory, or paired associates. Studies using only composite scores were excluded if the composite score included a measure not satisfying the EM criteria above; and (5) Study design of any length, frequency, duration, volume, or intensity, but acute exercise interventions were excluded. Study design must have isolated the effects of the aerobic group from the control group.

Search strategy. A literature search was conducted on April 1st, 2021, using the following databases: PubMed, CINAHL, EMBASE, PsycINFO, and CENTRAL. The following combination of search terms was used: memory OR recall OR verbal learning test OR reminding test OR story recall OR list learning OR word list OR paired associations AND randomized control trial\* OR clinical trial OR RCT AND exercise<sup>\*</sup> OR physical activity OR physical training AND aerobic AND older adult\* OR aging OR aged OR elderly OR geriatric OR cognitive impairment OR cognitive decline OR memory decline (see Supplementary Data 1 for database-specific syntax). We included only English-language articles. All articles published before April 1st, 2021 were included. We identified additional titles by a manual search of relevant journals (International Journal of Sports Medicine, Frontiers in Aging Neuroscience, and British Journal of Sports Medicine) and by identifying references in the six aforementioned meta-analyses. All returned titles were uploaded to Rayyan QCRI, a web application for systematic reviews<sup>41</sup>. The first author screened all returned titles to exclude duplicate studies. The first and second authors independently screened the remaining 699 titles and abstracts for eligibility based on the inclusion criteria described above. The authors agreed on the eligibility decision for 673 articles (96%); the first author resolved disagreements on the remaining 26. If articles appeared to be eligible but did not provide enough data to estimate an effect size or enough information about the intervention, the first author contacted the study's corresponding author (k = 9). Eight of the contacted authors were able to provide the requested data (k = 8). One of these studies included both MCI and dementia patients, and the author provided the requested data for only the patients with MCI42. Six studies included in previous meta-analyses were excluded from our analysis for the following reasons: (1) did not include a measure primarily evaluating EM15,43, (2) EM composite score included a measure that was not primarily evaluating EM<sup>23</sup>, (3) AE group included strength training<sup>24,44</sup>, and 4) did not include a control group<sup>45</sup>. In sum, 36 studies measured EM following an AE intervention and were included in the meta-analysis (Fig. 1).

**Data extraction**. All data were extracted and coded by the first author (see Supplementary Table 1 for a full list of variables and Supplementary Data 2 for the source data underlying all analyses, figures, and tables). All extracted variables were reviewed by the second author to ensure data accuracy. Where available, the pre and post mean, the pre and post standard deviation, and the sample size for the exercise and control groups at baseline (a conservative intent-to-treat framework) were considered in the calculation of effect size. When these data were not available, change from baseline, the standard deviation of the mean change, and the number of participants at each assessment for the exercise and control groups was extracted. Consistent with previous meta-analyses<sup>30</sup>, we assumed a moderate (0.5) correlation between baseline and post-intervention values. Improvement in EM performance was coded as a positive change score.

#### Data coding

Sample characteristics. Sex distribution, cognitive status, and age were coded into categorical variables. Sex distribution was coded using a median split: low (0-64% female) or high (65-100% female). Cognitive status was coded into one of two categories normal or impaired—using the study-defined diagnosis. Mean age was coded using a median split: young-old (55-68 years old) or old-old (69-85 years old). For a list of studies and their coded variables of interest, see Supplementary Data 3.

Intervention characteristics. Control group type, reporting of prescribed intensity, intervention length, session duration, session frequency, and intervention volume were coded into categorical or ordinal variables. The possibility of finding effects of an AE intervention relative to control is likely influenced by features of the control group. Thus, we coded control group type into three categories: no-contact (standard medical practice or wait-list), inactive (inactive control condition not involving any form of physical activity, such as education, painting, and social interaction), or physically active (involving physical activity without an aerobic component, such as stretching). Due to the infrequency of studies adequately reporting the prescribed intensity of the AE intervention, we categorized prescribed intensity simply as reported or not reported. Intervention length was coded as number of weeks and categorized into tertiles based on the distribution of included studies: short (6-17 weeks), medium (18-39 weeks), or long (40-65 weeks). When intervention length was reported in months, a conversion to weeks was calculated using the following criteria: 3 months = 13 weeks, 4 months = 17 weeks, 6 months=26 weeks, and 12 months = 52 weeks. Session duration was coded as the maximum number of minutes per exercise session, including any time spent warming up and cooling down. Session duration was coded using a median split: short (15-45 min) or long (50-90 min). Session frequency was coded into tertiles based on the distribution of included studies: low (1-2 sessions/week), medium (3 sessions/week), and high (4-7 sessions/week). When a study required a certain number of in-person supervised sessions/week and encouraged participants to exercise at home during the week, the number of in-person supervised sessions was coded. Intervention volume



Fig. 1 Prisma flow diagram. Adapted from: Moher D, Liberati A, Tetzlaff J, Altman DG, The PRISMA Group (2009). Preferred Reporting Items for Systematic Reviews and Meta-Analyses: The PRISMA Statement. PLoS Med 6(7): e1000097. doi:10.1371/journal.pmed1000097

was operationalized as total training minutes and calculated as the product of the intervention length, frequency, and duration. Intervention volume was coded into tertiles based on the distribution of included studies: low (<2100 total minutes), medium (2100–3900 total minutes), or high (>3900 total minutes).

**Quality assessment.** The Physiotherapy Evidence Database (PEDro) scale was used as a template to assess study quality (Supplementary Table 2)<sup>46</sup>. This scale has been used extensively in the evaluation of methodology of similar studies, and it is a valid and reliable quality assessment tool<sup>47–50</sup>. Study quality was calculated by adding the total number of criteria met on the checklist (possible range = 0–11). The first author assessed the quality of each study, and the second author reviewed the scores to ensure accuracy. Although PEDro does not provide specific instructions for classifying studies, the following criteria were used in accordance with previous meta-analyses:<sup>50,51</sup> scores 0–3 indicate poor-quality, scores 4–5 indicate fair-quality, and scores 6–11 indicate good- to excellent-quality.

**Analyses.** Comprehensive Meta-Analysis software (CMA; Version 3) was used to transform the standard mean difference into Hedges' g and to calculate the variance and 95% confidence interval (CI) for each study's effect size<sup>52</sup>. In studies with multiple treatment groups, we extracted data only for AE or control groups, excluding other intervention groups, such as those with cognitive training or resistance training. The individual effect size were then combined to create a summary effect size using the inverse variance of the individual effect size as weights.

Therefore, studies did not contribute equally to the summary effect size; rather, studies with greater variance (less precision) in their effect size estimate contributed less to the summary effect size than studies with smaller variance (greater precision). The summary effect size was calculated using a random-effects model, which assumes that each study has a different true effect size; we expected different true effect sizes for each study since studies were not matched on sample characteristics (e.g., age, cognitive status) or study design (e.g., control group type, dose-parameters, adherence) that could influence the magnitude of the effect size<sup>52</sup>.

For Aim 1, CMA was used to calculate the overall mean effect size of AE on EM by averaging across all measures of EM. Some studies reported means for multiple measures of EM. In these instances, we averaged the effect sizes across the measures and components when estimating the mean effect size for that study. Two studies produced two publications each that examined different measures of EM that were combined for the purpose of this analysis<sup>25,53-55</sup>. In addition, one study produced three publications examining overlapping and different measures of EM using different subsamples; the overlapping measure was included once using the largest subsample and the remaining measures were included as separate studies for the purpose of this analysis<sup>39,40,56</sup>. A leave-one-out analysis was performed, where each study was left out sequentially, to address whether a specific study was influential in the effects of AE on EM. Supplementary analyses were performed to calculate the overall mean effect size of AE on memory scores (total recall across learning trials, immediate recall, delayed recall, or recognition) and task performance (word list or story recall).

We tested for heterogeneity in the effect size by calculating the *Q*-statistic. A *Q*-statistic with a *p* value < 0.05 suggests significant heterogeneity in effect sizes. Significant heterogeneity indicates that the true effect size differs between studies. If significant heterogeneity was detected, an  $I^2$  statistic was then calculated to estimate the percentage of the total variability in effect sizes across studies due to true heterogeneity.

For Aim 2, we used CMA to conduct mixed-effects subgroup analyses and meta-regression analyses to investigate the studyspecific variables that might moderate the strength of the effect size across studies. Moderation effects were analyzed using the following categorical moderators: sex distribution, cognitive status, age, control group type, reported prescribed intensity, intervention length, session duration, session frequency, and intervention volume. First, the mixed-effects model was used to estimate the within-subgroup effect sizes for each moderator. Then, moderating effects were tested using Q-statistics to test the heterogeneity between effect sizes of subgroups defined by each moderator. The  $R^2$  statistic, which denotes the proportion of the true variance explained by the moderator, was obtained from the meta-regression model as the effect size of the moderator. Finally, the estimation of Hedges' g and its corresponding test for the zero effect were conducted stratified by subgroup. All analyses were conducted at the significance level of p < 0.05.

Sensitivity analyses. We performed a sensitivity analysis to investigate the impact of study quality. It was expected that higher-quality studies would provide more accurate effect size estimates. Therefore, we re-ran the primary analysis to examine whether any association changed when only high-quality studies were included (study quality score  $\geq 6$ ). If meta-analytic findings were primarily driven by poorer-quality studies, then we would be more cautious when interpreting results. A sensitivity analysis was also performed for sex distribution using a cut-off of 50% (<50% female or ≥50% female). Sensitivity analyses were also performed to assess whether a pre-post correlation value of 0 or 0.9 impacted the results. A sensitivity analysis was also performed to assess whether removing one publication with a mean age less than 60 years impacted the results<sup>57</sup>. Lastly, a sensitivity analysis was conducted to assess whether removing the three publications that included sessions in the anaerobic zone impacted the results<sup>39,40,56</sup>.

Risk of bias. Publication bias was assessed by a funnel plot, Duval and Tweedie's Trim and Fill, Egger's unweighted regression asymmetry test, and the Copas selection method. Trim and Fill uses an iterative process to remove the most extreme studies with small sample sizes, recompute an unbiased estimate of the effect size, and create a funnel plot that includes the observed studies and the imputed studies. If the shift of the effect size is small, we can be more confident that the reported effect is valid. A p value of Egger's asymmetry test > 0.05 indicates a symmetrical distribution of effect sizes and a low risk of publication bias. The Copas selection model gradually increases the positive association between the precision of a study and the probability that it is included in the meta-analysis until the test for residual selection bias is nonsignificant<sup>58,59</sup>. The Copas selection model was conducted at the significance level of 0.1 to test for residual selection bias using the R package metasens<sup>60</sup>.

Statistical power. A retrospective power calculation was conducted for the moderating effects using the *power*.*analysis.subgroup* function from the R package dmetar<sup>61</sup>. We also computed the sample size required for a future RCT to achieve 80% power for a two-sided *t*-test at the significance level of p = 0.05.

#### Results

Study characteristics. The initial search yielded a total of 1278 potentially relevant studies (Fig. 1). Of these, 36 RCTs met all inclusion criteria, representing data from 2750 participants (see Supplementary Data 4 for a list of studies). All included studies provided information about our moderators of interest and were published between April 1985-March 2021. Study sample sizes across exercise and control groups ranged from 15-389 participants, with a mean sample size per group per study of 38. The average age ranged from 59 to 85 years, with a sample-sizeweighted mean age of 70.82 years. Mean age was similar across exercise (70.64) and control groups (71.01). Females comprised 66.40% of the participants across all studies. Twenty studies included individuals with NC and 16 studies included individuals with impaired cognition, consisting of 1 SCD and 15 MCI studies. One study involved a placebo in both the aerobic and control groups<sup>35</sup>.

Effects of aerobic exercise on episodic memory. Figure 2 presents a summary of the effects of AE on EM using data from 36 RCTs. Consistent with our hypothesis, we found that AE enhanced EM (g [95% CI] = 0.28 [0.10-0.46]; p = 0.002). Given the observed effect size and CI, 404 participants (202 per intervention group) are needed in future studies to achieve 80% statistical power. Sensitivity analyses revealed similar results when a pre-post correlation value of 0 (p = 0.005) or 0.9 (p < 0.001) was used. Sensitivity analyses revealed similar results when excluding the three publications that included anaerobic sessions. Sensitivity analyses also revealed similar results when excluding one publication with a mean age less than 60 years. When removing each study in turn with a leave-one-out cross-validation analysis, all g's remained >0.17. and all *p*'s remained < 0.005. Supplementary analyses revealed a significant positive effect of AE on all memory scores, except learning and recognition, and all tasks (Supplementary Figs. 1-6).

We found significant heterogeneity in the full sample (Q(df) = 170.00 (35); p < 0.001), such that 79.41% of the variation in effect sizes could be attributed to true differences in the effect size across studies. Egger's unweighted regression asymmetry test suggests potential publication bias (t(34) = 2.07; p = 0.023). The Trim and Fill analysis suggests that if we remove the asymmetric studies, the effect size would increase to g [95% CI] = 0.38 [0.20-0.56] (Fig. 3). Using the Copas selection model, the adjusted effect size is smaller than the original effect size and no longer statistically significant (g [95% CI] = 0.10[-0.04, 0.24]; p = 0.16). Further, the Copas selection model revealed an estimated probability of publishing the trial with the largest standard error of 0.71 and an approximated number of unpublished studies of 14.

**Moderation analyses.** We tested whether the following sample and intervention characteristics moderated the strength of the effect size across studies: sex distribution, cognitive status, age, control group type, reported prescribed intensity, intervention length, session duration, session frequency, or intervention volume (Table 2). The results of the meta-regression analysis suggested that only age acted as a significant moderator (*Q*-statistic = 4.92; p = 0.027). The effect size was moderately small ( $R^2 = 0.09$ ). The effect sizes for the other moderators were small and did not provide statistical evidence of a moderating effect at p < 0.05.

Mixed-effects subgroup analyses revealed a significant effect for studies with a high percentage of females (g = 0.34; p = 0.009) but not those with a low percentage of females (g = 0.23; p = 0.086). Similar results were found when using 50% as the cut-off. These

Study Name	Statistics for each study								_	Hedges's g and 95% C			
	Hedges's	Standard error	Variance	Lower limit	Upper limit	Z-Value	p-Value						
Antunes 2015a	1.221	0.342	0.117	0.550	1.892	3.566	0.000		- I	<u> </u>	<b></b>	- I	
Antunes, 2015b	0.991	0.313	0.098	0.376	1.605	3.160	0.002				<u> </u>		
Blumenthal, 1989	-0.075	0.239	0.057	-0.544	0.394	-0.312	0.755				·		
Blumenthal, 2020	-0.339	0.225	0.050	-0.779	0.102	-1.508	0.132						
Cherup 2018	0.377	0.492	0.242	-0.587	1.341	0 766	0 444				_		
Fabre 2002	1,435	0.564	0.318	0.330	2,540	2.546	0.011			· · ·			
Ferreira 2015	-0.201	0.297	0.088	-0 784	0.381	-0.677	0.499						
Gaitán 2019	0.434	0.407	0.166	-0.364	1,233	1.066	0.286				_		
Hassmén, 1997	0.791	0.349	0.122	0.106	1.476	2.265	0.024						
liuin 2013	0.053	0.245	0.060	-0.428	0.534	0.216	0.829					1	
Jonasson, 2017	0.236	0.260	0.068	-0.274	0.747	0.908	0.364			- <b>f</b> +			
Kramer, 2001	2,727	0.257	0.066	2.223	3.232	10.600	0.000					_ I	
Lam. 2012	0,171	0,102	0.010	-0.030	0.371	1.671	0.095						
Lam 2015	-0 177	0 120	0.014	-0.412	0.059	-1 470	0 141						
Lazarou 2017	0.357	0.177	0.031	0.011	0 703	2 0 2 1	0.043						
Legault 2011	0 107	0.327	0 107	-0.534	0 747	0.326	0744						
Maass 2015	-0.264	0.312	0.097	-0.876	0.347	-0.847	0.397						
Maki 2012	0.054	0 163	0.026	-0.264	0.373	0 334	0.738			_			
Matura 2017	-0.060	0.272	0.074	-0.594	0 474	-0.219	0.826						
Morris 2017	0.072	0.281	0.079	-0.479	0.624	0.257	0.797						
Mortimer 2012	0.116	0.255	0.065	-0 384	0.616	0.453	0.651			i			
Nanamateu 2012	0.258	0.261	0.068	-0.253	0.768	0.400	0.322						
Oken 2006	0.000	0.208	0.043	-0.408	0.408	0.000	1 000						
Perri 1985	0.342	0.306	0.094	-0.259	0.943	1 116	0.265				-		
Poinsatte 2019	1 142	0.517	0.268	0.128	2 156	2 206	0.027						
Ruscheweyh 2011	0.465	0.311	0.097	-0 144	1 074	1 4 9 6	0.135						
Scherder 2005	-0.033	0.356	0 127	-0.731	0.665	-0.092	0.927			<u> </u>			
Stuckenschneider 2020	0.131	0.183	0.034	-0.228	0.490	0.718	0.473					1	
Tarumi 2019	-0.340	0.242	0.058	-0.814	0.133	-1 409	0.159			F			
ten Brinke, 2015	0.172	0.260	0.067	-0.337	0.681	0.663	0.507			- <b>- - -</b>			
Thomas 2020	0.652	0.365	0 133	-0.064	1.368	1 786	0.074			, i i i i i i i i i i i i i i i i i i i			
van Uffelen, 2008	-0.157	0.162	0.026	-0.474	0.160	-0.968	0.333			`			
Varela 2018	0.216	0.318	0 101	-0.407	0.839	0.679	0.497			_ <u>_</u>			
Vidoni 2021	-0.031	0.195	0.038	-0.413	0.351	-0 158	0.874					1	
Yoney-Seligmann 2021	-0 118	0.374	0 140	-0.851	0.615	-0 317	0.752						
7hu 2018	0.903	0.268	0.072	0.378	1 4 2 8	3 370	0.001				- I	1	
210, 2010	0.281	0.091	0.008	0.104	0.459	3.108	0.002			-			
	0.201		0.000			5.100		-3.60	-1.80	0.00	1.80	3.60	
									Favors Control		Favors Aerobic		





Funnel Plot of Standard Error by Hedges's g

**Fig. 3 Funnel plot of standard error by Hedges' g with imputed studies.** The observed studies (*n* = 36) are shown as open circles and the observed Hedges' g is shown as an open diamond at 0.28 [0.10-0.46]. The five imputed studies are shown as filled circles and the imputed Hedges' g is shown as a filled diamond at 0.38 [0.20-0.56].

subgroup analyses also indicated a significant effect for NC (g=0.42; p=0.001) but not for participants with impaired cognition (g=0.14; p=0.300). In line with the moderator analyses described above, there was a significant effect for studies with young-old participants (g=0.49; p<0.001) but not for old-old participants (g=0.10; p=0.418). The type of control group also influenced the results; there was a difference in EM when studies used a no-contact (g=0.38; p=0.018) or physically active (i.e., stretching) (g=0.35; p=0.028) control group but not when studies used an inactive control group (i.e., education, social

interaction) (g = 0.12; p = 0.456). Interestingly, the effect of AE on EM was significant for studies that reported their prescribed intensity (g = 0.36; p = 0.001) and not for studies that did not report prescribed intensity (g = 0.10; p = 0.587). Features of the intervention also appeared to influence the effects. Specifically, there was a significant effect for studies of medium length (g = 0.46; p = 0.002) but not for short (g = 0.25; p = 0.140) or long (g = 0.13; p = 0.405) interventions. There was an effect for studies with both short (g = 0.32; p = 0.021) and long (g = 0.26; p = 0.040) session durations. Studies with medium frequency

#### Table 2 Moderation Analyses.

Moderator	No. of studies	Hedges' g	Standard error	Lower limit	Upper limit	p-value	Q-statistic	p-value <sup>†</sup>	<b>R</b> <sup>2</sup>
Sex Distribution (percent)							0.34	0.558	<0.01
Low (0-64)	18	0.23	0.14	-0.03	0.5	0.086			
High (65-100)	18	0.34	0.13	0.09	0.6	0.009			
All	36	0.29	0.09	0.1	0.47	0.002			
Cognitive Status							2.43	0.119	0.04
Impaired	16	0.14	0.13	-0.12	0.39	0.3			
Normal	20	0.42	0.13	0.17	0.66	0.001			
All	36	0.29	0.09	0.11	0.46	0.002			
Age (years)							4.92	0.027	0.09
Young-Old (55-68)	18	0.49	0.13	0.24	0.75	<.001			
Old-Old (69-75)	18	0.1	0.12	-0.14	0.34	0.418			
All	36	0.28	0.09	0.11	0.46	0.001			
Control Group Type							1.55	0.461	<0.01
No-Contact	12	0.38	0.16	0.07	0.7	0.018			
Inactive	12	0.12	0.16	-0.20	0.44	0.456			
Physically Active	12	0.35	0.16	0.04	0.67	0.028			
All	36	0.29	0.09	0.11	0.47	0.002			
Intensity Reporting							1.59	0.208	<0.01
Not Reported	9	0.1	0.18	-0.25	0.45	0.587			
Reported	27	0.36	0.11	0.15	0.57	0.001			
All	36	0.29	0.09	0.1	0.47	0.002			
Intervention Length (weeks)							2.3	0.316	0.01
Short (6–17)	12	0.25	0.17	-0.08	0.57	0.14			
Medium (18-39)	13	0.46	0.15	0.16	0.76	0.002			
Long (40-65)	11	0.13	0.16	-0.18	0.44	0.405			
All	36	0.29	0.09	0.11	0.47	0.002			
Session Duration (min)							0.09	0.76	<0.01
Short (15-45)	17	0.32	0.14	0.05	0.59	0.021			
Long (50-90)	19	0.26	0.13	0.01	0.51	0.04			
All	36	0.29	0.09	0.1	0.47	0.002			
Session Frequency (sessions/week)							1.58	0.453	<0.01
Low (1-2)	11	0.14	0.16	-0.18	0.45	0.387			
Medium (3)	20	0.39	0.13	0.14	0.64	0.002			
High (4-7)	5	0.23	0.26	-0.27	0.74	0.369			
All	36	0.29	0.09	0.11	0.47	0.002			
Intervention Volume (total min)							0.19	0.909	<0.01
Low (<2100)	12	0.23	0.17	-0.11	0.57	0.188			
Medium (2100-3900)	12	0.31	0.16	-0.01	0.63	0.06			
High (>3900)	12	0.33	0.17	0.004	0.65	0.047			
All	36	0.29	0.1	0.1	0.48	0.003			

(g = 0.39; p = 0.002) had a significant effect, but studies with low (g = 0.14; p = 0.387) or high (g = 0.23; p = 0.369) session frequency did not. Finally, there was a significant effect for studies with a high volume of prescribed AE (g = 0.33; p = 0.047) but not for studies with a low (g = 0.23; p = 0.188) or medium (g = 0.31; p = 0.060) volume.

**Quality assessment**. Twenty-eight of the 36 studies were judged to be of high methodological quality because they scored  $\geq 6$  on the 11-item quality criteria. Supplementary Data 3 presents details on the methodological quality of each study. A post hoc analysis revealed a significant positive effect of AE on EM when only high-quality studies were included (*g* [95% CI] = 0.18 [0.03-0.32]; *p* = 0.016).

#### Discussion

In accordance with Aim 1, our meta-analysis of adults  $\geq$ 55 years without dementia demonstrates that participation in AE improves EM. In accordance with Aim 2, we found that AE improved EM in studies with NC, those with a mean age of 55–68, and those above the median in female participants. We also found that the EM exercise effect was only significant in

studies with a physically active or no-contact control group, that reported a prescribed AE intensity, and with session durations lasting 15–90 min three sessions/week for 18–39 weeks to achieve >3900 total minutes of activity. These observations have high practical relevance for the use of AE as a low-cost, accessible, non-pharmaceutical intervention to improve EM in late adulthood.

Previous meta-analyses had reported that AE does not influence EM among older adults with and without cognitive impairment<sup>28–32</sup>. The discrepancy in findings between this metaanalysis and previous meta-analyses may be due to methodological differences. Prior meta-analyses had restricted inclusion criteria, leading to a smaller number of included studies with AE and EM outcomes (e.g.,  $N_{\text{Northey}} = 9$ ). We included all studies of participants without dementia, which permitted us to include an additional 20 RCTs not included in previous meta-analyses. Further, another strength of this meta-analysis is the strict EM inclusion criteria, excluding studies with a measure not primarily targeting EM, such as spatial working memory, that was previously included in meta-analyses. By collapsing across various EM tasks that exhibit differential task demands on the hippocampus or recruit heterogeneous anatomical correlates, previous meta-analyses may have reduced their sensitivity to detect an effect.

Many studies included in the analyses did not reveal significant AE effects when considered individually, likely due to insufficient power because of small sample sizes. The power analysis revealed that 404 participants (202 per group) are needed in future studies to achieve 80% statistical power. None of the studies included in the meta-analysis had a sample size of this magnitude; the largest sample size was  $389^{62}$ , which gave 78.75% power to detect an effect size of g = 0.28, but studies on average had only 38 participants per group. The evidence for an AE effect became much stronger when all the data were pooled since most effects were in the positive direction.

Supplementary analyses revealed a significant positive effect of AE on immediate recall and delayed recall scores but not learning or recognition scores. There is research to suggest that conventional memory networks may not be sufficient or important for total learning performance and interpreting learning scores as episodic memory may be inaccurate<sup>63</sup>. These results are also consistent with the long-standing finding in cognitive psychology that age differences are smaller for recognition than recall<sup>64</sup>, such that there may be fewer age-related deficits that need to be mitigated. The literature suggests that these age differences in recall and recognition may be due to a common underlying process rather than ceiling effects in recognition tasks, such that older adults rely more on a general feeling of familiarity and are impaired in recollecting specific events, which is more important in recall tasks<sup>64</sup>. Further, the underlying process of familiarity is thought to rely on a more diffuse network of cortical and subcortical regions than the underlying process of recollection, which is closely linked with the hippocampus<sup>65</sup>. Thus, these regions associated with familiarity may be less modifiable with AE. It cannot be ruled out that the type of recognition score (i.e., hits, false alarms, d', percent accuracy) could have influenced the results, as most studies used percent accuracy or did not specify the measure used. We also found a significant positive effect of AE on word-list and story-recall tasks. Due to an insufficient number of studies examining face/name recognition, object memory, or paired associates, we were not able to examine or compare the effect of AE on these dimensions of EM. We were also not able to examine differences between studies that assess the "what," "where," or "when" components of EM given that most studies employed word list recall (75%) or story recall (33%) tasks, which tap the "what" component. We did not include spatial working memory or constructional praxis tasks due to their recruitment of heterogeneous anatomical correlates outside of the hippocampus. Thus, we were not able to distinguish effects on more distinct aspects of EM, such as visuospatial memory, or compare verbal memory to visuospatial memory. Future studies would benefit from comparing verbal memory with visuospatial memory to help provide information about the specificity of AE effects.

We had sufficient power to detect moderating effects of age, so any other possible moderators should be interpreted with caution. Moderation analyses revealed that age was a significant moderator of the effects of AE on EM. One explanation for our results is that there may be an opportune window during which AE has a more profound effect on EM. In particular, the larger benefits of AE for the young-old group may result from age differences in general health; comorbid conditions, medications, safety concerns, and other neuropathological issues, which are often more prevalent in older populations, could negatively impact adherence or the efficacy of the intervention. However, our results do not allow us to determine whether AE earlier in the lifespan can change the trajectory of cognitive aging because the current metaanalysis focused on adults with a mean age  $\geq$ 55 years at baseline. To address this gap, future studies would benefit from examining a younger population and following them across time to assess how AE at a single time point impacts EM later in life.

Although we did not find significant moderation as a function of sex distribution, subgroup analyses demonstrated that AE had favorable effects among studies composed mostly of female participants. These sex effects may be driven by differences in encoding strategies, such that females tend to utilize strategies (i.e., semantic clustering, word rhymes) that are associated with better performance<sup>66</sup>. It is possible that AE enhances the utilization or effectiveness of these strategies during encoding. These sex effects may also be driven by differences in exercise efficacy. Numerous studies have shown greater effects of AE for females compared to males on measures of executive functioning and information processing speed<sup>34,35</sup>. The sex differences may be related to sex steroid hormones estradiol and testosterone<sup>29</sup>. Extensive literature supports the role of sex hormones in neuroplasticity and the preservation of cognitive function, with evidence suggesting that greater circulating levels of estradiol and testosterone after the onset of menopause and andropause are associated with better EM in females but not males<sup>67,68</sup>. During development and reproductive years, the hormonal environment has lasting effects on brain structure, brain function, muscles, adipose tissue, and other organs<sup>69,70</sup>. Further, acute and chronic exercise increase sex steroid levels in the brain, muscles, and circulation<sup>71</sup>. Thus, even in late adulthood, exercise influences the brain and organs in a sex-dependent manner.

Subgroup analyses demonstrated that AE had favorable effects among participants with NC but not among participants with impaired cognition. There are several potential reasons for null results in impaired populations. Adherence to interventions varies greatly among studies of patients with MCI due to practical, psychological, and cognitive factors<sup>72</sup>. Additionally, given that our categorization of normal and impaired was based on study-defined diagnoses, there was significant heterogeneity across studies regarding diagnostic criteria-for example, one study<sup>73</sup> defined normal cognition as  $\geq 24$  on the Mini-Mental State Examination (MMSE), while another study used the same criteria to define MCI<sup>35</sup>, and yet another study defined MCI as a score of 7 or higher on the 12-item version of the MMSE<sup>27</sup>. This variability in diagnostic criteria may have hindered our ability to detect a significant effect among patients with impaired cognition. Lastly, although higher levels of physical activity are associated with reduced disease progression in patients with MCI<sup>74</sup>, the magnitude of the benefits may be smaller among individuals who have already developed neuropathology. Our results suggest that AE may be maximally beneficial among adults ≥55 years who are sedentary before any changes in cognition are detected clinically.

Although control group type was not a significant moderator, subgroup analyses demonstrated that AE had favorable effects among studies with a no-contact or physically active control group (i.e., stretching) but not among studies with an inactive control group (i.e., education). These finding suggest that while physical activity without a primary aerobic component is insufficient to promote significant changes in EM, education and social interactions may have a marginal effect, and, thus, may dampen the ability to detect significant effects of AE. Education or social interaction control groups may have promoted changes in lifestyle habits that had measurable benefits on EM. However, future research is needed to further explore potential underlying mechanisms.

While prescribed intensity was not a significant moderator, subgroup analyses demonstrated that AE had favorable effects only among studies that reported prescribed intervention intensity. Studies that report prescribed intensity may be better controlled, such that participants are monitored to ensure the target intensity is reached. It is also possible that studies that report prescribed intensity individualize the prescription for each participant based on their pre-training aerobic capacity. Given the heterogeneity across studies in reporting prescribed intervention intensity and the lack of variability among studies that reported prescribed intensity, we were not able to examine whether variation in intensity was a significant moderator of the effects of AE on EM.

Although intervention length, duration, frequency, and volume were not significant moderators, subgroup analyses demonstrated that AE had favorable effects among studies 18-39 weeks long with three sessions/week lasting 15-90 min each, achieving >3900 total minutes of activity. Given these results, the public health recommended target of 150 min/week of moderate-intensity physical activity<sup>75</sup> can be reached in a variety of ways. For example, 3900 min of activity could include 50-minute sessions 3 days a week for 26 weeks. Our results show that interventions 6-17 or 40-65 weeks, interventions with 1-2or 4–7 sessions/week, and interventions prescribing  $\leq$ 3900 total training minutes were not robust enough to result in significant differences in EM. It is possible that the most intense interventions-those 40-65 weeks or 4-7 sessions/week - yielded null results because interventions with greater time commitments could have had several limitations minimizing their observed effects, including greater subject burden, decreased adherence, increased attrition and withdrawal, and less precise measurement of exercise adoption. However, given the heterogeneity across studies regarding reporting these values, we could not examine whether variation in rates of adherence or attrition significantly moderated the effects described here. Further, intervention length and frequency varied substantially even among studies with >3900 total minutes of activity; interventions ranged from 26 to 65 weeks long with 2-7 sessions/week. This limits our ability to definitively state an optimal exercise dose at which EM benefits are detected or strengthened. Nonetheless, our results indicate that the public health recommendations of 150 min/week would require interventions greater than 26 weeks to detect EM improvements among adults  $\geq$ 55 years.

There are likely many behavioral and cellular mechanisms by which AE influences EM. For example, moderate-intensity exercise interventions improve sleep quality, which has a critical role in memory consolidation<sup>76,77</sup>. Exercise also improves mood, which can positively impact memory, although direct tests these possible mechanisms from RCTs remain of inconclusive<sup>78-80</sup>. There are also many physiological changes following exercise, including decreased body weight<sup>81</sup> and cardiovascular responses and reactivity<sup>82</sup>, which could influence brain health outcomes, including EM. EM is dependent on hippocampal function<sup>83</sup>, and exercise appears to positively impact hippocampal structure and function, especially in older adult populations<sup>15-17</sup>. On a cellular level, AE increases the expression and secretion of numerous growth factors and neurotrophins in the hippocampus, such as brain-derived neurotrophic factor (BDNF), which in turn enhances synaptic plasticity, cell proliferation, and survival<sup>84-86</sup>. In humans, an AErelated increase in hippocampal volume was associated with increased circulating levels of BDNF<sup>15</sup>. Additionally, AE might regulate an anti-inflammatory response in humans by modulating anti-inflammatory and proinflammatory cytokines<sup>87</sup>, and in rats, this change in cytokine levels in the hippocampus reduced hippocampal-related memory decline<sup>88</sup>. However, the behavioral and cellular transducers of the benefits of AE in humans are not fully understood and should be further investigated to optimally prescribe therapeutic interventions.

This literature is marked by several methodological limitations. The limited ethnic and racial diversity across samples restricts the generalizability of these results. Further, intervention intensity was often not reported and limited in variability, so a more sensitive classification could not be used in this review. The included studies also did not consistently report supervision, change in cardiorespiratory fitness, withdrawal/attrition, and attendance/adherence, limiting our ability to systematically examine whether these key intervention characteristics impact the effect of AE on EM. Future studies would benefit from reporting these aspects of the fidelity of the intervention so meta-analyses can clearly summarize the impact of these measures. Finally, studies varied in the way missing data was handled, with some studies using an intent-to-treat approach and others using a perprotocol approach depending on the purpose of the study. Studies using a per-protocol approach may have produced biases that may have overestimated the effect. In addition, there is evidence of publication bias and asymmetry; while the more conservative Copas selection model revealed a smaller effect size after accounting for the effects of small studies, the Trim and Fill suggests an effect size significantly larger than 0. Further, although we followed the PRISMA procedure for risk of bias, there are limitations of PRISMA and the PEDro scale that could influence the results from this meta-analysis (e.g., the summary score and cutoff do not acknowledge that some criterions are more critical to the validity of a study than others). Many of the included studies did not report EM performance for males and females separately or young-old and old-old participants separately, requiring us to categorize each study into one of two subgroups. However, future studies would benefit from reporting the effect for each group separately (e.g., males versus females) to allow for a more appropriate examination of these moderators using effect sizes for each group. In addition, we did not have access to individual participant data, which would have allowed for a more complex analysis of the impact of the moderators on EM. However, evidence suggests that group analyses are adequate and even outperform individual analyses in situations with a small number of trials per condition<sup>89</sup>. Further, even individuallevel data are essentially grouped data, as they are aggregated across multiple stimulus items<sup>90</sup>. Given that many of the moderators had small effect sizes, future research is needed to further validate the trends that appeared in these data, as well as other moderators that may be important.

Future research would also benefit from more long-term follow-up after the completion of the intervention to assess whether the benefits persist and whether they could delay progression to MCI or dementia. In addition, future studies must examine which type of training can be prescribed to promote the greatest benefits on EM. This may include multimodal training or resistance training, supervised or unsupervised training, group or individual training, and low-, moderate- or high-intensity training. Finally, while the effect of AE on EM was small and required a large sample size to detect a significant effect, a greater focus on middle-aged adults would allow for an investigation of whether benefits of AE on EM are more observable before the age of 55.

AE positively influences EM among adults  $\geq$ 55 years without dementia, with larger effects observed among various sample and intervention characteristics – the clearest moderator being age. These results suggest that EM should be routinely monitored starting at age 55 or earlier so that interventions can be implemented before changes in cognition become clinically apparent. The subgroup results could inform practitioners about how AE can maximally benefit EM in this age range. These findings highlight the importance for health professionals to communicate

the many benefits of exercise for brain health before the appearance of memory problems.

**Reporting summary**. Further information on research design is available in the Nature Research Reporting Summary linked to this article.

#### **Data availability**

Source data underlying all analyses, figures, and tables are available in Supplementary Data 2.

Received: 23 July 2021; Accepted: 18 January 2022; Published online: 17 February 2022

#### References

- Tulving, E. Episodic memory: from mind to brain. Annu. Rev. Psychol. 53, 1–25 (2002).
- Hodges, J., Tulving, E. & Craik, F. Memory in the dementias. in *The Oxford Handbook of Memory* 441–459 (Oxford University Press, 2000).
- Rajan, K. et al. Population estimate of people with clinical AD and mild cognitive impairment in the United States (2020–2060). *Alzheimers Dement*. 17, 1966–1975 (2021).
- 4. He, W., Goodkind, D. & Kowal, P. An Aging World: 2015. (2016).
- Madan, C. R. Rethinking the definition of episodic memory. Can. J. Exp. Psychol. 74, 183–192 (2020).
- Eichenbaum, H. Prefrontal-hippocampal interactions in episodic memory. Nat. Rev. Neurosci. 18, 547–558 (2017).
- DiNapoli, E. A., Wu, B. & Scogin, F. Social isolation and cognitive function in appalachian older adults. *Res. Aging* 36, 161–179 (2014).
- Tomaszewski Farias, S. et al. Longitudinal changes in memory and executive functioning are associated with longitudinal change in instrumental activities of daily living in older adults. *Clin. Neuropsychol.* 23, 446–461 (2009).
- Taylor, C. A., Bouldin, E. D. & McGuire, L. C. Subjective cognitive decline among adults aged ≥45 years—United States, 2015–2016. *Morb. Mortal Wkly Rep.* 67, 753–757 (2018).
- Raz, N. et al. Regional brain changes in aging healthy adults: general trends, individual differences and modifiers. *Cereb. Cortex* 15, 1676–1689 (2005).
- Jessen, F. et al. A conceptual framework for research on subjective cognitive decline in preclinical Alzheimer's disease. *Alzheimer's Dement.* 10, 844–852 (2014).
- Reijs, B. L. R. et al. Association between later life lifestyle factors and alzheimer's disease biomarkers in non-demented individuals: a longitudinal descriptive cohort study. J. Alzheimers Dis. 60, 1387–1395 (2017).
- Hillman, C. H., Erickson, K. I. & Kramer, A. F. Be smart, exercise your heart: exercise effects on brain and cognition. *Nat. Rev. Neurosci.* 9, 58–65 (2008).
- Caspersen, C. J., Powell, K. E. & Christenson, G. M. Physical activity, exercise, and physical fitness: definitions and distinctions for health-related research. *Public Health Rep.* 100, 126–131 (1985).
- Erickson et al. Exercise training increases size of hippocampus and improves memory. PNAS 108, 3017–3022 (2011).
- Maass, A. et al. Vascular hippocampal plasticity after aerobic exercise in older adults. *Mol. Psychiatry* 20, 585–593 (2015).
- Voss, M. W. et al. Plasticity of brain networks in a randomized intervention trial of exercise training in older adults. Front. Aging Neurosci. 2, 32 (2010).
- 18. Andrews-Hanna, J. R. et al. Disruption of large-scale brain systems in advanced aging. *Neuron* **56**, 924–935 (2007).
- Buckley, R. F. et al. Functional network integrity presages cognitive decline in preclinical Alzheimer disease. *Neurology* 89, 29–37 (2017).
- Wilckens, K. A. et al. Exercise interventions preserve hippocampal volume: a meta-analysis. *Hippocampus* 31, 335–347 (2021).
- van Praag, H., Christie, B. R., Sejnowski, T. J. & Gage, F. H. Running enhances neurogenesis, learning, and long-term potentiation in mice. *Proc. Natl. Acad. Sci. U.S.A.* 96, 13427–13431 (1999).
- Vaynman, S., Ying, Z. & Gomez-Pinilla, F. Hippocampal BDNF mediates the efficacy of exercise on synaptic plasticity and cognition. *Eur. J. Neurosci.* 20, 2580–2590 (2004).
- Komulainen, P. et al. Exercise, fitness and cognition—a randomised controlled trial in older individuals: The DR's EXTRA study. *Eur. Geriatr. Med.* 1, 266–272 (2010).
- Lautenschlager, N. T. et al. Effect of physical activity on cognitive function in older adults at risk for Alzheimer disease: a randomized trial. *JAMA* 300, 1027–1037 (2008).

- 25. Nagamatsu, L. S. et al. Physical activity improves verbal and spatial memory in older adults with probable mild cognitive impairment: a 6-month randomized controlled trial. *J. Aging Res.* **2013**, 861893 (2013).
- Lam, L. C. W., Chan, W. C., Leung, T., Fung, A. W.-T. & Leung, E. M.-F. Would older adults with mild cognitive impairment adhere to and benefit from a structured lifestyle activity intervention to enhance cognition? A cluster randomized controlled trial. *PLoS One* **10**, e0118173 (2015).
- Scherder, E. J. A. et al. Physical activity and executive functions in the elderly with mild cognitive impairment. *Aging Ment. Health* 9, 272–280 (2005).
- Angevaren, M., Aufdemkampe, G., Verhaar, H. J. J., Aleman, A. & Vanhees, L. Physical activity and enhanced fitness to improve cognitive function in older people without known cognitive impairment. *Cochrane Database Syst. Rev.* CD005381 https://doi.org/10.1002/14651858.CD005381.pub3 (2008).
- Barha, C. K., Davis, J. C., Falck, R. S., Nagamatsu, L. S. & Liu-Ambrose, T. Sex differences in exercise efficacy to improve cognition: a systematic review and meta-analysis of randomized controlled trials in older humans. *Front. Neuroendocrinol.* 46, 71–85 (2017).
- Northey, J. M., Cherbuin, N., Pumpa, K. L., Smee, D. J. & Rattray, B. Exercise interventions for cognitive function in adults older than 50: a systematic review with meta-analysis. *Br. J. Sports Med.* 52, 154–160 (2018).
- Song, D., Yu, D. S. F., Li, P. W. C. & Lei, Y. The effectiveness of physical exercise on cognitive and psychological outcomes in individuals with mild cognitive impairment: a systematic review and meta-analysis. *Int. J. Nurs. Stud.* **79**, 155–164 (2018).
- Young, J., Angevaren, M., Rusted, J. & Tabet, N. Aerobic exercise to improve cognitive function in older people without known cognitive impairment. *Cochrane Database Syst. Rev.* CD005381 https://doi.org/10.1002/ 14651858.CD005381.pub4 (2015).
- Ludyga, S., Gerber, M., Pühse, U., Looser, V. N. & Kamijo, K. Systematic review and meta-analysis investigating moderators of long-term effects of exercise on cognition in healthy individuals. *Nat. Hum. Behav.* 4, 603–612 (2020).
- Baker, L. D. et al. Effects of aerobic exercise on mild cognitive impairment: a controlled trial. Arch. Neurol. 67, 71–79 (2010).
- van Uffelen, J. G. Z., Chinapaw, M. J. M., van Mechelen, W. & Hopman-Rock, M. Walking or vitamin B for cognition in older adults with mild cognitive impairment? A randomised controlled trial. *Br. J. Sports Med.* 42, 344–351 (2008).
- de Frias, C. M., Nilsson, L.-G. & Herlitz, A. Sex differences in cognition are stable over a 10-year period in adulthood and old age. *Neuropsychol. Dev. Cogn. B Aging Neuropsychol. Cogn.* 13, 574–587 (2006).
- Federal Interagency Forum on Aging-Related Statistics. Older Americans 2012: Key Indicators of Wellbeing. 114–115 https://doi.org/10.1080/ 15398285.2013.756760 (2012).
- Moher, D., Liberati, A., Tetzlaff, J. & Altman, D. G. & PRISMA Group. Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement. *BMJ* 339, b2535 (2009).
- Thomas, B. P. et al. Brain perfusion change in patients with mild cognitive impairment after 12 months of aerobic exercise training. J. Alzheimers Dis. 75, 617–631 (2020).
- Poinsatte, K. et al. T and B cell subsets differentially correlate with amyloid deposition and neurocognitive function in patients with amnestic mild cognitive impairment after one year of physical activity. *Exerc. Immunol. Rev.* 25, 34–49 (2019).
- 41. Ouzzani, M., Hammady, H., Fedorowicz, Z. & Elmagarmid, A. Rayyan-a web and mobile app for systematic reviews. *Syst. Rev.* 5, 210 (2016).
- 42. Morris, J. K. et al. Aerobic exercise for Alzheimer's disease: a randomized controlled pilot trial. *PLoS One* **12**, e0170547 (2017).
- Moul, J., Goldman, B. & Warren, B. Physical activity and cognitive performance in the older population. J. Aging Phys. Activity 3, 135–145 (1995).
- 44. Barnes, D. E. et al. The Mental Activity and eXercise (MAX) trial: a randomized controlled trial to enhance cognitive function in older adults. *JAMA Intern. Med.* **173**, 797–804 (2013).
- Stevenson, J. S. & Topp, R. Effects of moderate and low intensity long-term exercise by older adults. *Res. Nurs. Health* 13, 209–218 (1990).
- Physiotherapy Evidence Database. PEDro scale. PEDro https://pedro.org.au/ (2021).
- de Morton, N. A. The PEDro scale is a valid measure of the methodological quality of clinical trials: a demographic study. *Aust. J. Physiother.* 55, 129–133 (2009).
- Maher, C. G., Sherrington, C., Herbert, R. D., Moseley, A. M. & Elkins, M. Reliability of the PEDro scale for rating quality of randomized controlled trials. *Phys. Ther.* 83, 713–721 (2003).
- Roig, M., Nordbrandt, S., Geertsen, S. S. & Nielsen, J. B. The effects of cardiovascular exercise on human memory: a review with meta-analysis. *Neurosci. Biobehav. Rev.* 37, 1645–1666 (2013).
- Sanders, L. M. J., Hortobágyi, T., la Bastide-van Gemert, S., van der Zee, E. A. & van Heuvelen, M. J. G. Dose-response relationship between exercise and cognitive function in older adults with and without cognitive impairment: a systematic review and meta-analysis. *PLoS One* 14, e0210036 (2019).

- Roig, M. et al. The effects of eccentric versus concentric resistance training on muscle strength and mass in healthy adults: a systematic review with metaanalysis. *Br. J. Sports Med.* 43, 556–568 (2009).
- Borenstein, M., Hedges, L. V., Higgins, J. P. & Rothstein, H. R. Comprehensive Meta-Anlaysis. (Biostat Inc., 2013).
- 53. Blumenthal, J. A. et al. Lifestyle and neurocognition in older adults with cognitive impairments: a randomized trial. *Neurology* **92**, e212–e223 (2019).
- Blumenthal, J. A. et al. Longer term effects of diet and exercise on neurocognition: 1-year follow-up of the ENLIGHTEN trial. J. Am. Geriatr. Soc. 68, 559–568 (2020).
- Nagamatsu, L. S., Handy, T. C., Hsu, C. L., Voss, M. & Liu-Ambrose, T. Resistance training promotes cognitive and functional brain plasticity in seniors with probable mild cognitive impairment. *Arch. Intern. Med.* 172, 666–668 (2012).
- 56. Tarumi, T. et al. Exercise training in amnestic mild cognitive impairment: a one-year randomized controlled trial. J. Alzheimers Dis. 71, 421–433 (2019).
- 57. Ruscheweyh, R. et al. Physical activity and memory functions: an interventional study. *Neurobiol. Aging* **32**, 1304–1319 (2011).
- Copas, J. B. & Shi, J. Q. A sensitivity analysis for publication bias in systematic reviews. Stat. Methods Med. Res. 10, 251–265 (2001).
- Schwarzer, G., Carpenter, J. & Rücker, G. Empirical evaluation suggests Copas selection model preferable to trim-and-fill method for selection bias in metaanalysis. J. Clin. Epidemiol. 63, 282–288 (2010).
- 60. Schwarzer, G. & Carpenter, J. metasens: Statistical Methods for Sensitivity Analysis in Meta-Analysis. (2021).
- Harrer, M., Cuijpers, P., Furukawa, T. & Ebert, D. D. dmetar: Companion R Package For The Guide 'Doing Meta-Analysis in R'. (2019).
- Lam, L. C. W. et al. A 1-year randomized controlled trial comparing mind body exercise (Tai Chi) with stretching and toning exercise on cognitive function in older Chinese adults at risk of cognitive decline. *J. Am. Med. Dir. Assoc.* 13, 568.e15–20 (2012).
- Casaletto, K. B. et al. Is "Learning" episodic memory? Distinct cognitive and neuroanatomic correlates of immediate recall during learning trials in neurologically normal aging and neurodegenerative cohorts. *Neuropsychologia* 102, 19–28 (2017).
- Rhodes, S., Greene, N. R. & Naveh-Benjamin, M. Age-related differences in recall and recognition: a meta-analysis. *Psychon. Bull. Rev.* 26, 1529–1547 (2019).
- Eichenbaum, H., Yonelinas, A. P. & Ranganath, C. The medial temporal lobe and recognition memory. *Annu. Rev. Neurosci.* 30, 123–152 (2007).
- 66. Loprinzi, P. D. & Frith, E. The role of sex in memory function: considerations and recommendations in the context of exercise. J. Clin. Med. 7, 132 (2018).
- Wolf, O. T. & Kirschbaum, C. Endogenous estradiol and testosterone levels are associated with cognitive performance in older women and men. *Horm. Behav.* 41, 259–266 (2002).
- Hamson, D. K., Roes, M. M. & Galea, L. A. M. Sex hormones and cognition: neuroendocrine influences on memory and learning. *Compr. Physiol.* 6, 1295–1337 (2016).
- 69. de Vries, G. J. & Forger, N. G. Sex differences in the brain: a whole body perspective. *Biol. Sex Differ.* 6, 15 (2015).
- Arnold, A. P. The organizational-activational hypothesis as the foundation for a unified theory of sexual differentiation of all mammalian tissues. *Horm. Behav.* 55, 570–578 (2009).
- Janssen, J. A. M. J. L. Impact of physical exercise on endocrine aging. Front. Horm. Res. 47, 68–81 (2016).
- 72. van der Wardt, V. et al. Adherence support strategies for exercise interventions in people with mild cognitive impairment and dementia: a systematic review. *Prev. Med. Rep.* **7**, 38–45 (2017).
- 73. Antunes, H. K. et al. Effects of a physical fitness program on memory and blood viscosity in sedentary elderly men. *Braz. J. Med. Biol. Res.* 48, 805–812 (2015).
- 74. Burns, J. M. et al. Cardiorespiratory fitness and brain atrophy in early Alzheimer disease. *Neurology* **71**, 210–216 (2008).
- 2018 Physical Activity Guidelines Advisory Committee. 2018 Physical Activity Guidelines Advisory Committee Scientific Report. (2018).
- King, A. C., Oman, R. F., Brassington, G. S., Bliwise, D. L. & Haskell, W. L. Moderate-intensity exercise and self-rated quality of sleep in older adults. a randomized controlled trial. *JAMA* 277, 32–37 (1997).
- Walker, M. P. The role of sleep in cognition and emotion. *Ann. N.Y. Acad. Sci.* 1156, 168–197 (2009).
- Lichtenberg, P. A., Ross, T., Millis, S. R. & Manning, C. A. The relationship between depression and cognition in older adults: a cross-validation study. J. Gerontol. B Psychol. Sci. Soc. Sci. 50, P25–P32 (1995).
- Penedo, F. J. & Dahn, J. R. Exercise and well-being: a review of mental and physical health benefits associated with physical activity. *Curr. Opin. Psychiatry* 18, 189–193 (2005).
- Stillman, C. M., Cohen, J., Lehman, M. E. & Erickson, K. I. Mediators of physical activity on neurocognitive function: a review at multiple levels of analysis. *Front. Hum. Neurosci.* 10, 626 (2016).

- Buccola, V. A. & Stone, W. J. Effects of jogging and cycling programs on physiological and personality variables in aged men. *Res. Q.* 46, 134–139 (1975).
- Stamford, B. A. Physiological effects of training upon institutionalized geriatric men. J. Gerontol. 27, 451–455 (1972).
- 83. Knierim, J. J. The hippocampus. Curr. Biol. 25, R1116-R1121 (2015).
- Liu, P. & Nusslock, R. Exercise-mediated neurogenesis in the hippocampus via BDNF. Front. Neurosci. 12, 52 (2018).
- Rossi, C. et al. Brain-derived neurotrophic factor (BDNF) is required for the enhancement of hippocampal neurogenesis following environmental enrichment. *Eur. J. Neurosci.* 24, 1850–1856 (2006).
- van Praag, H., Shubert, T., Zhao, C. & Gage, F. H. Exercise enhances learning and hippocampal neurogenesis in aged mice. *J. Neurosci.* 25, 8680–8685 (2005).
- Woods, J. A., Wilund, K. R., Martin, S. A. & Kistler, B. M. Exercise, inflammation and aging. *Aging Dis.* 3, 130–140 (2012).
- Lovatel, G. A. et al. Treadmill exercise induces age-related changes in aversive memory, neuroinflammatory and epigenetic processes in the rat hippocampus. *Neurobiol. Learn. Mem.* **101**, 94–102 (2013).
- Cohen, A. L., Sanborn, A. N. & Shiffrin, R. M. Model evaluation using grouped or individual data. *Psychon. Bull. Rev.* 15, 692–712 (2008).
- Pratte, M. S. & Rouder, J. N. Assessing the dissociability of recollection and familiarity in recognition memory. J. Exp. Psychol. Learn. Mem. Cogn. 38, 1591–1607 (2012).

#### Acknowledgements

This research was supported by a grant awarded from the National Institutes of Health to S.L.A. (T32GM081760).

#### Author contributions

S.L.A. developed the study design, conducted a literature search, extracted the data, ran the analyses, and wrote the manuscript. T.B. assisted with data extraction and quality assessment. C.K. assisted with the analyses and writing the manuscript. X.Z. assisted with the analyses and writing the manuscript. S.D.D. supervised the analyses and assisted with writing the manuscript. S.D.D. supervised the analyses and assisted with the development of the study design, supervised the analyses, and assisted with the development of the study design, supervised the analyses, and assisted with the development of the study design, supervised the analyses, and assisted with the development of the study design, supervised the analyses, and assisted with writing the manuscript. M.W.V. assisted with the development of the study design, supervised the analyses, and assisted with writing the manuscript. S.H.F. supervised the analyses and assisted with writing the manuscript. K.I.E. assisted with the development of the study design, supervised the analyses, and assisted with writing the manuscript. K.I.E. assisted with the development of the study design, supervised the analyses, and assisted with writing the manuscript. S.H.F. supervised the analyses and assisted with writing the manuscript.

#### **Competing interests**

The authors declare no competing interests.

#### **Additional information**

**Supplementary information** The online version contains supplementary material available at https://doi.org/10.1038/s43856-022-00079-7.

Correspondence and requests for materials should be addressed to Sarah L. Aghjayan.

Peer review information Communications Medicine thanks Terence Quinn and the other, anonymous, reviewer(s) for their contribution to the peer review of this work.

Reprints and permission information is available at http://www.nature.com/reprints

**Publisher's note** Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

**Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons license, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons license and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this license, visit http://creativecommons.org/ licenses/by/4.0/.

© The Author(s) 2022